Author's response to reviews

Title: Pretreatment of parecoxib attenuates hepatic ischemia reperfusion injury in rats.

Authors:

Tao Zhang (zhtao98@aliyun.com)
Yi Ma (369512714@qq.com)
Kang-Qing Xu (13719198392@139.com)
Wen-Qi Huang (huangwengimzk@163.com)

Version: 2 Date: 7 May 2015

Author's response to reviews: see over
May 7, 2015

Dear Editor,

Thank you for your consideration of our manuscript. We have made changes to the format of the paper and have provided point-by-point description of the changes made.

Description of the changes made

1) We have added the full name of the ethical committee and the guidelines we adhered to in this study to the manuscript.

2) We have provided the species and the number of the animals we used in this study in the manuscript.

3) We have provided the name of the animal center where we obtained the animals.

4) We have added a conclusion section in the main text of the manuscript.

5) We have explained the role of the funder in this study in the acknowledgements of the manuscript.

We hope that we have satisfactorily made the changes. Please do not hesitate to contact me if you have any further questions. Thank you very much.
Yours sincerely,

Wen-Qi Huang